Author's response to reviews

Title: Multidrug resistant Kluyvera ascorbata septicemia in an adult patient: a case report

Authors:

Shannon Moonah (smoonah@howard.edu)
Kavita Deonarine (kavitasoleil@yahoo.com)
Clyde Freeman (clcfreeman@Howard.edu)

Version: 2 Date: 3 December 2009

Author's response to reviews:

This case was initially sent to the Cases Journal. BioMed Central will be ceasing the publication of the journal, hence the transfer to JMCR.

We believe this case contributes significantly to evidence based medical practice, for example if there is an adult patient with a multidrug resistant Kluyvera ascorbata isolated from the blood, we provide the only clinical evidence to support sucessful treatment with a carbapenem.